Author's response to reviews

Title: Association of the rs6982567 near GDF6 with neovascular age-related macular degeneration and polypoidal choroidal vasculopathy in a Han Chinese cohort

Authors:

Yuying Ji (jillea@163.com)
Xiongze Zhang (xiongze.zhang@gmail.com)
Kunfang Wu (kf.1012.k@163.com)
Yu Su (sy_daisy1206@qq.com)
Meng Li (drlimeng@163.com)
Chengguo Zuo (chengguozuo@163.com)
Feng Wen (wenfeng208@foxmail.com)

Version: 2
Date: 23 July 2014

Author's response to reviews: see over
Dear editor,

Thank you for your suggestions.

The research was carried out within an appropriate ethical framework. It followed the tenets of the Declaration of Helsinki and the protocol was approved by the institutional review board at the Zhongshan Ophthalmic Center of Sun Yat-sen University (lot number: KYNL015). All the participants assigned the written informed consent after explanation of the purpose and process of the study. These were mentioned in line 82-84 on Page 4.

The naming of the genes mentioned in this article was reported in a standard way. It could be found in the “HUGO Gene Nomenclature Committee (HGNC)” database. We provided the complete information about the primer sequences used in this study, which were available for other researches. They were mentioned in line 113-115 on Page 5.

We hope that these revisions are satisfactory and thank you very much for your work on our paper.

Sincerely,

Feng Wen, M.D., Ph.D.,
State Key Laboratory of Ophthalmology,
Zhongshan Ophthalmic Center,
Sun Yat-sen University
54 South Xianlie Road, Guangzhou, 510060, China
Phone: +86 20 87330292
E-mail: wenfeng208@foxmail.com